Author's response to reviews

Title: A novel pathogenic MLH1 missense mutation, c.112A>C, p.Asn38His, in six families with Lynch syndrome

Authors:

Els van Riel (E.vanRiel@umcutrecht.nl)
Margreet GEM Ausems (M.G.E.M.Ausems@umcutrecht.nl)
Frans BL Hogervorst (f.hogervorst@nki.nl)
Irma Kluijt (i.kluijt@nki.nl)
Marielle E van Gijn (M.E.vanGijn@umcutrecht.nl)
Jeanne van Echtelt (J.vanEchtelt@umcutrecht.nl)
Karen Scheidel-Jacobse (k.scheidel@antoniusziekenhuis.nl)
Eric FAM Hennekam (F.A.M.Hennekam@umcutrecht.nl)
Rein P Stulp (r.p.stulp@medgen.umcg.nl)
Yvonne J Vos (y.j.vos@medgen.umcg.nl)
G Johan A Offerhaus (G.J.A.Offerhaus@umcutrecht.nl)
Fred H Menko (FH.Menko@vumc.nl)
Johan JP Gille (JJP.Gille@vumc.nl)

Version: 4 Date: 2 August 2010

Author's response to reviews: see over
Dear sir/madam,

Please find enclosed the revised manuscript “A novel pathogenic MLH1 missense mutation, c.112A>C, p.Asn38His, in six families with Lynch syndrome” by E. van Riel, MGEM Ausems, FBL Hogervorst, I Kluijt, ME van Gijn, J van Echtelt, K Scheidel-Jacobse, EFAM Hennekam, RP Stulp, YJ Vos, GJA Offerhaus, FH Menko and JJP Gille. The manuscript has been revised according to the comments of the independent editorial production team.

We would like to address the formatting changes made in the manuscript:

1. Figure legends: We have organized the figure legends section so that the appropriate key corresponds to the relevant figure. For each figure, a separate legend is provided.
2. The figures are cropped as closely as possible to minimise white space around the image. Legends from Figure 2 and 3 are removed from these figures and placed in the figure legends section of the revised manuscript.

We trust that this revised manuscript will meet the wishes of the editorial production team.

Yours sincerely, also on behalf of the co-authors,

Ms. dr. E. van Riel
Genetic Counselor

Visiting address:
Lundlaan 6
3584 EA Utrecht

Postal address:
Huispostnummer KC4.084.2
Postbus 85090
3508 AB Utrecht